Author's response to reviews

Title: Bcl11b mutations identified in murine lymphomas increase the proliferation rate of hematopoietic progenitor cells

Authors:

Anneli Karlsson (annka@ibk.liu.se)
Amanda Nordigarden (amano@ibk.liu.se)
Jan-Ingvar Jonsson (janjo@ibk.liu.se)
Peter Söderkvist (petso@ibk.liu.se)

Version: 5 Date: 13 October 2007

Author's response to reviews: see over
Dr. Sandra Le
Assistant Editor,
BMC Cancer

Re: 1488734051143192

Dear Dr. Le,

Please find attached the final version of the manuscript “Bcl11b mutations identified in murine lymphomas increase the proliferation rate of hematopoietic progenitor cells”. As requested, the following changes have been made:

1. Key words have been removed from the abstract page.
2. The headline “Declaration of competing interests” have been changed to “Competing interests”.
3. Figure legends contain no symbols.
4. Any blank pages have been removed.
5. The figures have been cropped to minimize the space around them.

We hope that you will find these changes satisfactory.

Sincerely,
Anneli Karlsson

Anneli Karlsson
Division of Cell Biology/Department of Biomedicine and Surgery
Faculty of Health Sciences, Linköping University
S-581 85 Linköping, Sweden
E-mail: annka@ibk.liu.se
Tel. +46 13 222679